Author's response to reviews

Title: Immigrant women's experiences of postpartum depression in Canada: a protocol for systematic review using a narrative synthesis

Authors:

Gina MA Higginbottom (gina.higginbottom@ualberta.ca)
Myfanwy Morgan (myfanwy.morgan@kcl.ac.uk)
Joyce O'Mahony (jomahony@mtroyal.ca)
Yvonne Chiu (yvonnechiu@shaw.ca)
Deb Kocay (debbie.kocay@phac-aspc.gc.ca)
Mirande Alexandre (misamalex@yahoo.ca)
Joan Forgeron (joan.forgeron@albertahealthservices.ca)
Marilyn Young (marilyn.young@albertahealthservices.ca)

Version: 3 Date: 24 July 2013

Author's response to reviews: see over
Dear Editors,

Thank you for your review of this manuscript. We have addressed the reviewer’s suggestions in the uploaded revised manuscript, and have provided a response below for our changes. We look forward to hearing from you soon.

Sincerely,

Gina Higginbottom

Response to reviewer comments:

1. Regarding the suggestion to consider using PRESS - Our search strategy was developed in collaboration with and will be implemented by a specialist librarian and knowledge scientist therefore we have not had the opportunity to include the PRESS principles in this review.
2. We have corrected the spelling error related to the “Cochrane Collaboration”